Author's response to reviews

Title: Comparing Parecoxib and Ketorolac as Preemptive Analgesia in Patients Undergoing Posterior Lumbar Spinal Fusion: A Prospective Randomized Double-Blinded Placebo-Controlled Trial

Authors:

Koopong Siribumrungwong (koopongs@hotmail.com)
Julin Cheewakidakarn (chulin_c@hotmail.com)
Boonsin Tangtrakulwanich (boonsin.b@psu.ac.th)
Sasikaan Nimmaanrat (snimmaanrat@yahoo.com)

Version: 5 Date: 23 February 2015

Author's response to reviews: see over
Cover letter

Dear Reviewer (Dr. Kitti Jirarattanaphochai)

There was no change from last comments.

Dear Reviewer (Dr. Surachai Sae-Jung)

1. Please consistent the wording: pre-emptive VS preemptive
   
   I already changed the wording to “preemptive”

2. Some repeated measurement (such as pain): please mention in statistical part or result that the repeated measure anova (for continuous data) or Friedman test (categorical data) had no statistical significance.
   
   I already mention repeated pain measurement in results part.

Sincerely,
Dr. Koopong Siribumrungwong
Department of Orthopaedics
Faculty of Medicine
Prince of Songkla University